Author's response to reviews

Title: Economic evaluation of three populational screening strategies for cervical cancer in the county of Valles Occidental: CRICERVA clinical trial.

Authors:

Amelia Acera (amelia.acera@gmail.com)
Ana Rodriguez (arodriguezma.mn.ics@gencat.cat)
Marta Traper (marta.traper-bertran@brunel.ac.uk)
Pilar Soteras (psoteras.mn.ics@gencat.cat)
Norman Sanchez (normansanchez@gencat.cat)
Josep Maria Bonet (jimbonets@gencat.cat)
Josep Maria Manresa (jmanresa@idiapjgol.info)
Pablo Hidalgo (cbarbera.mn.ics@gencat.cat)
Pere Toran (ptoran@idiapjgol.info)
Gemma Prieto (gpljenny@gmail.com)

Version: 2 Date: 22 June 2011

Author's response to reviews:

After sending the document, we are not sure to indicate correctly the journal: we want to send it to BMC Health Services Research.

Thank you very much